Author's response to reviews

Title: Evaluation of Anthelmintic potential of the Ethiopian Medicinal Plant Embelia schimperi Vatke in vivo and in vitro against some Intestinal Parasites

Authors:

Yared Debebe (yare9999@yahoo.com)
Mesfin Tefera (mesfin_07@yahoo.com)
Walelign Mekonnen (WalelignM@yahoo.com)
Dawit Abebe (dawisoul@gmail.com)
Samuel Woldekidan (Samuelwoldekidan@rocketmail.com)
Abiy Abebe (abiyaboq@yahoo.com)
Temesgen Menberu (temeadonay@gmail.com)
Yehualashet Belete (yehualashete.belete3@gmail.com)
Bethelhem Belayneh (betelhemb7@gmail.com)
Berhanu Tesfaye (rberhanu@yahoo.com)
Ibrahim Nasir (ibranaser@yahoo.com)
Kidist Yirsaw (Y.kidist@yahoo.com)
Hirut Basha (hirut_basha@yahoo.com)
Asrat Dawit (dawitasrat53@yahoo.com)
Asfaw Debella (asfawde bella@gmail.com)

Version: 2 Date: 23 December 2014

Author's response to reviews:

I have corrected my ethical statement confirming that we got ethical clearance for both collection of patients' stool samples and use of laboratory animals for experimental procedures from the scientific and ethical review office of Ethiopian Public Health institute. We also included we obtained written consent from patients.

I have also attached the written consent form at the end of the manuscript.